Author's response to reviews

Title: Reproductive desire in women with HIV infection in Spain, associated factors and motivations: A mixed-method study

Authors:

Victoria Hernando (vhernando@isciii.es)
Belen Alejos (balejos@isciii.es)
Debora Alvarez (dalvarez@isciii.es)
Marta Montero (martamontero72@gmail.com)
Maria Jesus Perez-Elias (mjperez90@gmail.com)
Jose Ramon Blanco (jrblanco@riojasalud.es)
Mar Masia (marmasiac@gmail.com)
Jorge del Romero (jorgedelromero@gmail.com)
Ignacio de los Santos (isantosg@hotmail.com)
Isabel Rio (isario@isciii.es)
Alicia Llacer (allacer@isciii.es)
and CoRIS (cohortecoris@gmail.com)

Version: 4 Date: 21 April 2014

Author's response to reviews: see over
Dear Editors,

We hereby resubmit our manuscript (reference number: MS1518036544115196) entitled, "Reproductive desire in women with HIV infection in Spain, associated factors and motivations: A mixed-method study", for consideration by the Editorial Board of the BMC Pregnancy & Childbirth as a Research Article.

The manuscript has been changed according to comments of the reviewers and the editor:

- E-mail addresses of all authors have been included in the Title Page
- A Conclusion Section has been included at the end of the main text
- A copy of the questionnaire used in our study has been added as additional file

We thank you again for giving us the opportunity to re-submit our work and look forward to your response.

Yours sincerely,

Victoria Hernando Sebastián
Centro Nacional de Epidemiología, Instituto de Salud Carlos III,
Avda. Monforte de Lemos, 5. 28029 Madrid. SPAIN.
Tel.: 34- 918222921
Fax: 34- 913877815
Email: vhernando@isciii.es